Author's response to reviews

**Title:** Joint effect of mitochondrial DNA 5178 C/A polymorphism and alcohol consumption on estimated glomerular filtration rate in male Japanese health check-up examinees: a cross-sectional study

**Authors:**

Akatsuki Kokaze (akokaze@med.showa-u.ac.jp)
Mamoru Ishikawa (mamoru.ishikawa@achs.jp)
Naomi Matsunaga (naomim@ks.kyorin-u.ac.jp)
Kanae Karita (kanaae@ks.kyorin-u.ac.jp)
Masao Yoshida (yohhy@ks.kyorin-u.ac.jp)
Naoki Shimada (naokismd@med.showa-u.ac.jp)
Tadahiro Ohtsu (tohtsu@med.showa-u.ac.jp)
Takako Shirasawa (shirasawa@med.showa-u.ac.jp)
Hirotaka Ochiai (h-ochiai@med.showa-u.ac.jp)
Hiromi Hoshino (hhiromi@med.showa-u.ac.jp)
Yutaka Takashima (yutakat@kyorin-u.ac.jp)

**Version:** 2  **Date:** 9 January 2013

**Author's response to reviews:** see over
Dr. Hayley Henderson,
Executive Editor, BioMed Central

January 9, 2013

Dear Dr. Henderson,

Thank you for your kind e-mail of June 13, 2012 regarding our manuscript (No. 9265848666631408).

We are sending the revised manuscript.
We hope that the revised manuscript is now suitable for publication of “BMC Nephrology”.

If you have any questions concerning the revised manuscript, please let us know by e-mail. We will be happy to answer them. We are looking forward to having a favorable response from you soon.

Yours sincerely,
Akatsuki KOKAZE M.D., Ph. D.
Department of Public Health, Showa University School of Medicine.